Author's response to reviews

Title: Sparing effects of selenium and ascorbic acid on vitamin C and E in guinea pig tissues

Authors:

Jesse Bertinato (jesse_bertinato@hc-sc.gc.ca)
Nick Hidiroglou (nick_hidiroglou@hc-sc.gc.ca)
Robert Peace (bob_peace@hc-sc.gc.ca)
Kevin A Cockell (kevin_cockell@hc-sc.gc.ca)
Keith D Trick (keith_trick@hc-sc.gc.ca)
Penny Jee (penny_jee@hc-sc.gc.ca)
Alex Giroux (alex_giroux@hc-sc.gc.ca)
Rene Madere (rene_madere@hc-sc.gc.ca)
Giuseppe Bonacci (joey_bonacci@hotmail.com)
Monica Iskandar (monica_iskandar@hc-sc.gc.ca)
Stephen Hayward (stephen_hayward@hc-sc.gc.ca)
Nicholas Giles (nicholas_giles@hc-sc.gc.ca)
Mary R L'Abbe (mary_l'abbe@hc-sc.gc.ca)

Version: 3 Date: 26 March 2007

Author's response to reviews:

The manuscript has been checked for topographical errors and the final version has been submitted.

Jesse Bertinato, Ph.D.
Research Scientist
HPFB, Nutrition Research Division
Health Canada, PL 2203C
Sir Frederick G. Banting Research Centre
251 Sir Frederick Banting Driveway
Ottawa, ON, Canada
K1A 0L2
Tel: (613) 957-0924
Fax: (613)941-6182
E-mail: jesse_bertinato@hc-sc.gc.ca